Author's response to reviews

Title: Knowledge brokering: Exploring the process of transferring knowledge into action

Authors:

Vicky L Ward (v.l.ward@leeds.ac.uk)
Allan O House (a.o.house@leeds.ac.uk)
Susan Hamer (s.hamer@leeds.ac.uk)

Version: 2 Date: 12 January 2009

Author's response to reviews: see over
Dear editors,

Find attached a revised version of our study protocol entitled ‘Knowledge brokering: exploring the process of transferring knowledge into action’.

In response to your comments:
(1) Document, within your manuscript, the name of the ethics committee which approved your study – this has been added to the manuscript under the heading ‘ethical approval’ and can be found on page 12.

(2) Document, within your manuscript, whether you will be receiving informed consent from your study participants and if this will be verbal or written – this has been added to the ‘participants’ section of the manuscript on page 9.

I trust that this has addressed your concerns and look forward to receiving details of publication dates very soon.

With best wishes
Dr Vicky Ward